Author’s response to reviews

Title: Isolated Palmar Dislocation of Distal Radioulnar Joint: A New Mechanism of Injury: A case report

Authors:

Xianke Lin (1515038@zju.edu.cn)

Hui Shen (shenhui33@sina.com)

Hui Lu (hitman1982@hotmail.com; huilu@zju.edu.cn)

Version: 2 Date: 19 Jul 2019

Author’s response to reviews:

Dear Prof. Amanpreet Athwal:

Thank you very much for your letter and advice. We have revised the manuscript, and would like to re-submit it for your consideration. The response to your comments are listed below this letter.

We hope that the revised version of the manuscript is now acceptable for publication in your journal.

I look forward to hearing from you soon.

With best wishes,

Yours sincerely, Hui Lu M.D.

Department of Orthopedics, The First Affiliated Hospital, College of Medicine, ZheJiang University. Huilu@zju.edu.cn
Replies to Assistant Editor

1. Title page

We noticed that the corresponding author information provided on your title page is different to the one you indicated in the editorial submission. Please amend one so that they match.

Answer: we make they match.

2. Funding

In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Answer: we change the section “…fund the collection and analysis of this study. ”

3. Authors' Contributions

Please remove the full name Hui Shen and replace it with the initials HS.

Answer: we replace it.

4. Formatting

-- Please rename your "Introduction" heading to "Background".

-- Please combine your Discussion and Conclusions sections into one, with a "Discussion and Conclusions" heading.

Answer: we make the changes.

5. Supplementary material

-- Please remove the letter to the editor and CARE checklist from the file inventory as these are no longer needed at this stage in the editorial process.

Answer: we remove them.

6. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Answer: we make the changes.